Author's response to reviews

Title: Variations in DEPDC5 gene and its association with chronic Hepatitis C virus infection in Saudi Arabia

Authors:

Mashael R Al-Anazi (maalanazi@kfshrc.edu.sa)
Sabine Matou Nasri (matouepnasrisa@NGHA.MED.SA)
Ayman A Abdo (aabdo@ksu.edu.sa)
Faisal M Sanai (faisalsanai@hotmail.com)
Mohammed Q Khan (MohamedKhan@kfshrc.edu.sa)
Ali Albenmousa (alibenmousa@gmail.com)
Hamad I Al-Ashgar (alashgar@kfshrc.edu.sa)
Nisreen Z Khalaf (nkhalaf35@kfshrc.edu.sa)
Mohammed N Al-Ahdal (ahdal@kfshrc.edu.sa)
Ahmed A Al-Qahtani (aqahtani@kfshrc.edu.sa)

Version: 6 Date: 20 October 2014

Author's response to reviews:

Author's response to editor

Title: Variations in DEPDC5 gene and its association with chronic Hepatitis C virus infection in Saudi Arabia

Authors:

Mashael R Al-Anazi, Sabine Matou Nasri, Ayman A Abdo, Faisal M Sanai, Mohammed Q Khan, Ali Albenmousa, Hamad I Al-Ashgar, Nisreen Z Khalaf, Mohammed N Al-Ahdal and Ahmed A Al-Qahtani

Version: 4 Date: 20 October 2014

Comment #1: Before we proceed to the next stages, please ensure that all the names of the IRB/ethics committee that approved the study are mentioned with the ethics approval statement. Please clarify within the manuscript whether it was written informed consent that was obtained from all participants (patients and health controls).

Approval of the institutional review boards (IRBs) of all participating hospitals and the written informed consent of all participating subjects were stated in the Materials and Methods section (Kindly see page 7, lines 157-159).

Comment #2: Please include a legend section for the additional files within the main manuscript document

Done
Comment #3: We would like to give you this opportunity to copyedit the manuscript and thoroughly check spelling and grammar to improve clarity and understanding.
Done